Author's response to reviews

Title: Randomised Phase I/II trial assessing the safety and efficacy of radiolabelled anti-carcinoembryonic antigen I131 KAb201 antibodies given intra-arterially or intravenously in patients with unresectable pancreatic adenocarcinoma

Authors:

Asma Sultana (mmasul@liv.ac.uk)
Sussanah Shore (ssshore73@liv.ac.uk)
Michael G T Raraty (mraraty@liverpool.ac.uk)
Sobhan Vinjamuri (Sobhan.Vinjamuri@rlbuht.nhs.uk)
Jonathan E Evans (Jonathan.Evans@rlbuht.nhs.uk)
Catrin Tudur Smith (cat1@liverpool.ac.uk)
Steven Lane (slane@liverpool.ac.uk)
Seema Chauhan (seema.chauhan@liv.ac.uk)
Lorraine Bosonnet (lorraine.bosonnet@rlbuht.nhs.uk)
Conall Garvey (Conall.Garvey@rlbuht.nhs.uk)
Robert Sutton (r.sutton@liv.ac.uk)
John P Neoptolemos (j.p.neoptolemos@liv.ac.uk)
Paula Ghaneh (p.ghaneh@liv.ac.uk)

Version: 2 Date: 30 October 2008

Author's response to reviews:

Dear Editor,

Thank you for your response. The paper has been revised in keeping with the points raised by the reviewers, and a detailed response to their comments are attached.

We await your response.

Thanking you,

Yours' sincerely,

Asma Sultana.